Contents lists available at ScienceDirect

Applied Energy

journal homepage: www.elsevier.com/locate/apenergy

Data-driven Urban Energy Simulation (DUE-S): A framework for integrating engineering simulation and machine learning methods in a multi-scale urban energy modeling workflow



AppliedEnergy

Alex Nutkiewicz, Zheng Yang, Rishee K. Jain*

Urban Informatics Lab, Dept. of Civil & Environmental Engineering, Stanford University, 473 Via Ortega, Stanford, CA 94305, USA

HIGHLIGHTS

- Introduces the need for urban building energy modeling and current approaches.
- Proposes a DUE-S framework that integrates machine learning and simulation methods.
- DUE-S models urban energy use on multiple spatial and temporal scales.
- · Evaluates the DUE-S framework on case study of 22 dense urban buildings.
- Achieves acceptable prediction accuracies at an urban scale.

ARTICLEINFO

Keywords: Building energy Data-driven Machine learning Multi-scale Simulation Urban energy modeling Urban context

ABSTRACT

The world is rapidly urbanizing, and the energy intensive built environment is becoming increasingly responsible for the world's energy consumption and associated environmental emissions. As a result, significant efforts have been put forth to develop methods that can accurately model and characterize building energy consumption in cities. These models aim to utilize physics-based building energy simulations, reduced-order calculations and statistical learning methods to assess the energy performance of buildings within a dense urban area. However, current urban building energy models are limited in their ability to account for the inter-building energy dynamics and urban microclimate factors that can have a substantial impact on building energy use. To overcome these limitations, this paper proposes a novel Data-driven Urban Energy Simulation (DUE-S) framework that integrates a network-based machine learning algorithm (ResNet) with engineering simulation to better understand how buildings consume energy on multiple temporal (hourly, daily, monthly) and spatial scales in a city (single building, block, urban). We validate the proposed DUE-S framework on a proof of concept case study of 22 densely located university buildings in California, USA. Our results indicate that the DUE-S framework is able to accurately predict urban scale energy consumption at hourly, daily and monthly intervals. Moreover, our results also demonstrate that the integration of data-driven and engineering simulation approaches can partially capture the inter-building energy dynamics and impacts of the urban context and merits future work to explore how they can be improved to predict sub-urban scale energy predictions (single building, block). In the end, successfully predicting and modeling the energy performance of urban buildings has the potential to inform the decision-making of a wide variety of urban sustainability stakeholders including architects, engineers and policymakers.

1. Introduction

The world is rapidly urbanizing. Over 50% of the world's population now resides in cities, and the number is expected to increase to 67% by 2050 [1]. Cities account for over 75% of all primary energy use and over 80% of greenhouse gas emissions, with the largest portion of such consumption (more than 40%) and related emissions coming from the built environment [2,3]. As a result, urban buildings represent a tremendous opportunity to enhance the energy sustainability of cities. According to recent estimates, as much as 90% of urban buildings are energy efficient, and up to 30% of an individual building's energy consumption is wasted [4].

* Corresponding author. *E-mail address:* rishee.jain@stanford.edu (R.K. Jain).

https://doi.org/10.1016/j.apenergy.2018.05.023



Received 3 January 2018; Received in revised form 3 May 2018; Accepted 5 May 2018 Available online 01 June 2018 0306-2619/ © 2018 Elsevier Ltd. All rights reserved.

Extensive academic and industrial efforts have been undertaken to develop energy conservation measures within individual buildings (e.g., demand driven heating/cooling controls). However, building energy use is significantly affected by other buildings (e.g., shading impacts heating and natural lighting) and microclimate factors (e.g., changes in wind patterns impact heat transfer and cooling loads). A key challenge in enhancing the energy efficiency of buildings in dense urban areas is the lack of accurate energy performance prediction models that consider this urban context. Current building energy models are limited in their ability to account for the inter-building energy dynamics and interdependencies that can have dynamic and non-linear impacts on the energy use of urban buildings. Without accurate performance characterization and prediction, designers and engineers struggle to assess the energy, environmenta, and economic implications of their early-stage design and retrofit decisions, thus failing to shape a building's energy use for its entire lifecycle. This challenge is further exacerbated by adjacent buildings and the overall urban area becoming increasingly energy intensive resulting in substantial energy, environmental and monetary impacts [5-7].

Rapid growth in new sensing technologies and emerging smart city initiatives has led to an explosion of structured and unstructured data streams describing buildings and their surrounding urban environment. Simultaneously, the field of artificial intelligence is quickly developing new machine learning models that harness these new data streams to predict and characterize a wide range of physical phenomeno a within cities (e.g., air pollution dynamics [8], traffic flow [9] and energy use [10]). The primary objective of this paper is to introduce a novel Datadriven Urban Energy Simulation (DUE-S) framework that aims to bridge the gap between traditional engineering-based energy simulation models and emerging data-driven machine learning models¹. We postulate that by integrating the two methods we can take the first step towards accurately characterizing the energy performance of urban buildings at multiple temporal (e.g., hourly, daily, monthly) and spatial (e.g., single building, block, urban) scales. These accurate characterizations can then help facilitate the assessment of design and retrofit decisions. The rest of the paper is organized as follows: Section 2 presents an overview of existing work on urban energy modeling and discusses the main gaps; Section 3 introduces the methodology of DUE-S that integrates a residual network machine learning model with engineering simulation to better understand how buildings consume energy on multiple temporal and spatial scales; Section 4 proposes the setup of a case study with 22 densely located university buildings in southern California, USA, and the measures used to validate the performance of DUE-S framework; Section 5 discusses the case study results; Section 6 outlines the limitations and future work; and Section 7 concludes the paper.

2. Background

Urban energy modeling is the virtual representation and reproduction of the energy performance for buildings located in an urban area. Generally, urban energy modeling aims to capture the urban context by simulating energy dynamics at multiple spatial and temporal scales. In this section, we provide a brief review of the existing literature on urban energy modeling: the surrounding urban context, multi-scale performance, and calibration. Finally, in order to contextualize our proposed model within the existing body of work we also provide a review of how emerging data-driven methods have been applied to the energy modeling problem setting.

2.1. Urban context

Energy is primarily consumed in buildings to smooth thermal loads (e.g., add or remove sensible and latent heat to achieve thermal balance with conduction, convection and radiation from outside and inside the building) and power loads (e.g., power lighting system, air handling equipment, computers, and other devices used by occupants). These loads are significantly influenced by the building's urban context through effects from neighboring buildings, vegetation and other urban systems [11–13]. For example, a building's outside temperature could be abnormally high due to urban heat island effects [14,15]. As the sun moves, the surrounding urban built environment may cast shadows and shadings that in turn impact a building's energy use [16]. Because air dynamically flows around and within buildings, wind is another key element that determines the rate of building heat transfer (e.g., convection), humidity, cooling and ventilation loads [17]. Wind speed and direction change drastically due to urban context, as nearby buildings and trees can influence wind patterns. Previous research indicates that fluid dynamics in an urban area should be included in energy modeling as it can have a substantial impact [18]. Furthermore, urban buildings can be served by district energy systems such as heating networks [19], district cooling plants [20] and energy hubs [21], making the energy use of one building highly interdependent on surrounding buildings. Lastly, the dynamics that occur in networks of building occupants can also impact urban energy use as connections and interactions among occupants have been found to vary the heating/cooling loads [22,23], lighting loads [24] and plug loads [25] across buildings.

Engineering simulation programs (e.g., EnergyPlus, DOE2, IES-VE) reproduce the physical energy processes of buildings by: (1) taking inputted building geometries and abstracting them to a network of connected nodes, (2) creating heat balance equations for all nodes across each hour of a virtual year and (3) solving those equations within each time step, using many assumed non-geometric building parameters to calculate a building's energy consumption. However, because there are a large number of nodes to model and equations to solve, simulating the performance of hundreds of buildings across a city at once is both time intensive and computationally expensive [26]. Efforts have been made to simplify this modeling process. Specifically, the geometries can be extracted from GIS (Geographic Information Systems) [27,28], CityGML [29], BIM (Building Information Modeling) [30], CAD (Computer Aided Design) [31] or digital images [32]. Additionally, non-geometric properties (e.g., building and construction material, operation schedules, HVAC systems) have been assumed based on "archetypes"-templates representing groups of buildings with similar properties-to reduce the number of input variables [26,30]. In order to define "archetypes," buildings are divided into groups based on properties like shape and age where buildings within each group are considered identical. While highly productive in reducing the amount of input variables into an energy simulation model, the characterization of "archetypes" is often ad-hoc and depends greatly on the availability of data [13]. As a result, it is often difficult to evaluate the reliability and authenticity of the results. New "hourglass" approaches [33,34] have begun to address some of the shortcomings of archetype-based models as they combine reductive archetype models with a re-diversification process in order to add stochastic variations to individual buildings and re-introduce diversity lost in the reductive archetype process. Moreover, reduced-order methods have also been developed to model urban energy use, including electrical circuit analogy based on resister-capacitor networks [35], energy demand calculations based on quasi-static monthly energy balance [36], degree-day estimations based on heat transfer coefficient [33], steady-state methods based on energy balance equations [19], thermal shoebox models based on insolation analysis and clustering [31] and reducedcomplexity models based on simplified state space methods [37]. However, such reduced-order methods often require large oversimplifications (e.g., a building is modeled as single thermal zone) [31]

 $^{^1}$ The short version of the paper was presented at ICAE2017, Aug 21–24, Cardiff, UK. This paper is a substantial extension of the short version of the conference paper.

and require making several strong assumptions (e.g., the heating set point is constant) [19].

Ongoing research to quantify the impact urban context has on building energy use has been primarily focused on modeling microclimatic factors alongside simulation. Several studies have quantified the effects radiation exchange has on building energy consumption. For example, ENVI-met, a three-dimensional microclimate model, has been integrated into EnergyPlus in order to quantify the interactive effects of CFD (Computational Fluid Dynamics) and thermodynamics on building energy consumption [38,39]. Other tools such as the Urban Modeling Interface (UMI) are focused on modeling conditions such as shading and urban heat island effects in addition to energy use within the context of urban planning [40]. So while new tools [16,41] are emerging that begin to incorporate some aspects of building interdependencies in dense urban areas and model the impact of energy retrofits, there remains a significant opportunity to develop new approaches to comprehensively capture all the influences of the urban context in building energy modeling.

2.2. Multi-scale performance

Accurately characterizing and modeling the energy performance of urban buildings on multiple scales is integral to gaining a comprehensive picture of the impact of early-stage design decisions and energy efficiency retrofit solutions. For example, insights from accurate building scale models can inform building operations and energy efficient building management [42]. Block scale modeling can reveal insights on how design changes (e.g., proposed building heights and orientation) impact solar shading on other buildings and facilitate block scale optimization of energy use [11]. Urban scale modeling can support decision-making regarding sustainable zoning and other planning guidelines that impact a city's overall morphology and subsequent energy use [43]. As a result, in order to maximize the use cases and facilitate more holistic sustainability and energy decision-making, it is integral that urban energy performance simulations are accurate on multiple scales.

Engineering modeling methods are powerful at simulating a single building with a standalone HVAC system [44] but fail when applied beyond a single building [45]. Even co-simulation, like the coupling of EnergyPlus and Modelica [20], faces difficulty in generating reliable simulation results at multiple scales. Similarly, reduced-order modeling methods are limited to the scale at which they are built and are not extensible to other scales. In order to overcome this issue, various methods for scaling single building simulations have been developed. These scaling methods include: using a single aggregated building to represent all individual buildings in a district [46], multiplying energy use of a single building by the number of buildings per archetype [26], scaling up results of an individual building model by a floor area weighting function [47] or adding up the energy use of individual buildings to represent the larger scale energy use [19]. While such scaling methods have enabled some modeling and characterization of urban energy use, they struggle to achieve highly accurate results at multiple scales simultaneously in a single comprehensive model. Even the prevailing bottom-up methods using a set of calibrated and validated archetypes [48] have limited capability to model energy use accurately at multiple scales as the numbers, characterization and representativeness of archetypes differ dramatically across scales.

Previous work [49] has shown that the high accuracy of an energy model at one scale does not guarantee high accuracy at others as the impacts of urban context are highly variable across scales, especially when the microclimate varies spatiotemporally. Obtaining high modeling accuracy at multiple scales simultaneously becomes increasingly difficult as the complexity exponentially increases due to the nonlinear interactions and interdependencies from thermal equilibrium, geometry, fluid dynamics, occupant networks and system responses within urban buildings. While previous work has demonstrated positive results at both district and building scales [19,50], such models have limited predictive ability to quantitatively estimate the impacts that design and/or retrofit changes in one building (e.g., new HVAC control strategy) have on the energy use of surrounding buildings. As a result, this paper aims to build upon previous work by proposing and validating an urban energy modeling framework that is capable of modeling energy performance at multiple scales (single building, block, urban).

2.3. Energy modeling calibration

Simulation is a context-related process, and results from uncalibrated models can deviate significantly (up to 90%) from actual energy use [51,52] - making calibration indispensable before any practical use can be made of the results. Urban energy modeling calibration is an inverse approximation as it requires the reconciling of simulation outputs with measured energy data. It is often overparameterized due to the large number of independent and interdependent input parameters and assumptions that must be specified. Because the inverse and over-parameterized characteristics of calibration could be naturally associated with misassumptions of building energy dynamics, they are some of the main drivers of uncertainty in the energy modeling process [53]. In order to overcome these issues, researchers [54,55] have utilized an iterative process to adjust input parameters and assumptions based on available data during the modeling process until the discrepancies between simulated results and measurements are within the ranges regulated by practice standards [56,57]. This genre of methods follows the trial-and-error principle or Bayesian update that requires significant time and effort as the performance of the energy model depends largely on ad-hoc data availability and subjective engineering judgment [58,59]. Another type of calibration focuses on applying statistical analysis to find the relationships between input parameters and actual energy use during the modeling process [60,61]. The input parameters can either be static (e.g., window-wall ratio) or dynamic (e.g., zone temperature). While this approach is computationally efficient and can enhance the accuracy of energy simulation models that consider the urban context [62], retraining with additional ground truth data is required if any changes are made to a building (e.g., building envelope retrofits), building system (e.g., HVAC control upgrades) or the surrounding urban environment (e.g., new buildings). More importantly, since the statistical relationships are constructed on a building-to-building basis and utilized to adjust simulation inputs, they can be overspecialized and may lack generalizability to other buildings and urban areas, and thus limit their applicability for modeling urban energy use at multiple scales.

2.4. Emerging data-driven approaches to urban energy modeling

The advent of smart meters and open data initiatives across major cities has made energy data available in varying spatial and temporal resolutions. This data availability combined with the rapid development of new data-driven machine learning models has spawned the use of computational intelligence to find hidden patterns of urban building energy use. As a result, these emerging data-driven energy models are able to achieve high degrees of accuracy and quantify the impacts of various energy covariates, such as building characteristics, weather, spatial patterns or use types [63]. Common models utilized in previous work to predict building energy consumption include multiple linear regression models [64], support vector regression [65], decision trees [66] and artificial neural networks [67]. Specifically, support vector regressions are kernel-based algorithms that excel when solving nonlinear problems and have been applied to model the energy use of multi-family residential buildings [10]. Decision trees are a subset of flexible algorithms that divide features into weighted branches to make predictions [66], and artificial neural networks are non-linear models that use interconnected neurons and activation functions to relate



Fig. 1. Proposed DUE-S modeling framework. Step 1 uses information from existing data sources to create individual building energy simulations. Step 2 employs a residual network model (ResNet) to predict the metered energy use by learning the uncertainties of the building energy simulations and their surrounding urban environment.

input, hidden and output layers of information [67]. Tso and Yao [68] compare the performance of decision trees, neural networks and multiple linear regression models to predict residential energy consumption in summer and winter seasons and find that both decision trees and neural networks perform well in this task. Unfortunately, the majority of these studies focuses exclusively on creating models to predict individual building energy use and have yet to explore how such methods extend to the urban context.

Recent work that has evaluated building energy use at the urban scale has been largely done for benchmarking or energy mapping purposes [69]. Clustering methods are used to divide large groups of buildings into sub-groups based on common features in order to evaluate them against a baseline building level [70]. Artificial neural networks have also been used on the multi-building scale to evaluate the energy consumption of schools and office buildings [71,72]. And while one study was able to apply ordinary least squares, random forest and support vector machines to predict building energy use with public data provided by energy disclosure policies, its results were limited because the temporal granularity of the data was only provided at the annual scale [73]. However, there is now increasing popularity for employing complex deep learning-based models that can take better advantage of the vast amounts of available data and improved computational power of GPUs and CPUs. Deep learning is the process of training neural networks, which are able to take a set of input features and figure out functions that map them to labeled predictions. Neural networks are modeled similarly to how neurons input, process and output information in the brain through a hierarchy of hidden layers. They are trained to perform a specific function by using a non-linear activation function to calibrate the weights of connections between neurons, also called "nodes," with the goal of minimizing a loss function, typically through gradient descent [74]. Deep learning has previously been applied to a wide variety of fields related to the urban built environment including forecasting solar and wind consumption [75], predicting traffic flows [76] and forecasting building energy consumption [77]. Specifically, convolutional neural networks (CNNs), which are characterized by their locally connected layers, have been used to represent the spatiotemporal nature of traffic flow [76]. CNNs have also been applied to individual building energy forecasting because of their ability to represent time series data in a grid topology [77]. In addition to CNNs, recurrent neural networks (RNNs) are also popular for time series forecasting because they use previous predictions as additional inputs for future ones. This type of model, for example, has been used to forecast wind speed for a network of wind stations [75].

While purely data-driven approaches are able to model the statistical relationships between building energy use and characteristics [67], they rely too much on the mathematical patterns in available data

without considering any underlying physics of building thermal and energy systems. Therefore, pure data-driven approaches lack the ability to accurately estimate the energy implications of early-stage design or new retrofits to buildings (e.g., installation of a new window material) that are not included in the training dataset. Since buildings in an urban area could have hundreds of possible design and/or retrofit changes, it is impossible to have a robust dataset for training an algorithm that encapsulates all potential options. As a result, this paper aims to bridge the gap between emerging data-driven machine learning and engineering energy simulation methods by proposing a Data-driven Urban Energy Simulation framework (DUE-S). DUE-S aims to integrate engineering simulation models with new data-driven machine learning models to enable the exploration of potential early design decisions and energy conservation measures while capturing the non-linear and complex interactions of the urban context. It is this novel integration of engineering and data-driven methods that we postulate enables the DUE-S model to accurately characterize and model the energy performance of buildings at multiple spatial and temporal scales. Furthermore, we clarify that our objective is not to propose an alternative for calibration of energy simulation model inputs with new data streams but to utilize a data-driven approach that learns the complex relationships and patterns of urban building energy use.

3. Methodology

In this section, we describe the mechanics of the DUE-S model—a two-step process that aims to integrate engineering-based and datadriven approaches, as shown in Fig. 1. This first step is to build baseline energy simulation models with an appropriate level of detail based on analysis of the tradeoffs between the modeling objective and availability of data (Step 1). The desired output of this step is periodic time series data for each building that captures its basic energy processes and use patterns. Subsequently, these time series become the structured data inputs to the residual network (ResNet) machine learning model (Step 2), where we use the machine learning algorithm to learn the relationships between the simulated and actual metered energy data for each building. The final output of the DUE-S model is an integrated urban energy model that can be utilized to predict and characterize the performance of buildings at multiple temporal and spatial scales.

3.1. Step 1: Baseline energy simulation models

Energy simulation is often over-parameterized and, if generated on a larger, urban scale, requires an infeasible amount of time and resources to accurately complete. The data available to create model inputs for each individual building varies drastically from building to



Fig. 2. Determination of model input based on the tradeoffs between the level of detail and data access.

building due to numerous factors ranging from the time period in which it was built to privacy concerns. As a result, it is important that a consistent level of detail across an entire target urban area should be determined for developing baseline energy simulation models in Step 1 of our proposed DUE-S framework. In order to do this, we first must determine the required level of detail and generate energy simulation models with at least this level of detail for each building in the urban study area.

In general, there is a tradeoff between the level of detail needed for an energy simulation and the resources needed to acquire the required supporting data. Specifically, the level of detail is associated with a model's performance and uncertainty. A higher level of detail could result in more accurate simulation results, however, could also require more precise building-related data. Different purposes of urban energy modeling require different levels of uncertainties from model input (e.g., the design of a new building envelope does not need detailed occupancy schedules as the building has not been occupied yet). The energy modeling objective is the critical value to determine the minimum level of detail (red arrow in Fig. 2). Details below the critical value are required at minimum to be actual in order to build the baseline energy models. On the other hand, the data access required to construct such details also needs to be considered. The scale of data availability regulated is represented in Fig. 2, where lower-level data is considered less reliable than higher-level data for determining input parameters and assumptions but is easier to collect [78]. The critical value (green arrow in Fig. 2) indicates the minimum level of data that is easily accessible to modelers (green zone in Fig. 2). As a result, the shaded grey region in Fig. 2 represents the model input determination that weighs the tradeoff between the level of detail required for the modeling objective (e.g., retrofit analysis, pre-design phase) and the availability of data (e.g., GIS shapefiles, sensor-based data) for developing models.

To be clear, the exact locations of critical values for the level of detail (red arrow in Fig. 2) and for data access (green arrow in Fig. 2) vary case by case, thus the final level of generalization of the energy model can differ. In this paper, it is assumed the objective of our urban energy model is to test the impacts of geometric designs and physical efficiency retrofits on building energy use in a dense urban area. Based on the analysis of tradeoffs between the level of detail input and data access, the baseline energy simulation models require the inputs of the following data primarily collected from publicly available datasets: weather, building shape, building height and number of floors. To model a site's climatic conditions, energy simulations often rely on Typical Meteorological Year (TMY) datasets, which contain the "most average" hourly weather information for a site, including variables such as dry bulb temperature, relative humidity, solar radiation and wind speed [79]. However, with the increasing availability of historical

weather data at the hourly interval [80], this can be substituted in place of a more generalized TMY file.

We use the building GIS shapefiles available from local municipal websites to construct building geometries. The data structure normally contains fields on building IDs, height, elevation and roofline area. By merging this information with the shapefiles in SketchUp, simplified massing models-often referred to as "2.5-D" models-can be constructed for each building to understand its area and orientation. The buildings can be further divided into floors using either additional information or calculations based on standard floor-to-floor height. Finally, the non-geometric building properties, which include the building's systems, schedules and internal constructions, must be defined. Since these fields are above the critical value of level of detail defined by our model's objective, we define input parameters based on the U.S. Department of Energy's (DOE) Commercial Reference Building models [81]. The DOE and several of its national laboratories used national data from the 2003 Commercial Building Energy Consumption Survey (CBECS) to determine an average mix of representative buildings. These reference models include 16 building types for 3 different construction periods and represent about 60% of the U.S. commercial building stock. Each of these models is available as a ZIP file with detailed spreadsheets of plug and process loads, construction assemblies, operating schedules and systems. In the case that certain non-geometric parameters are available for a specific urban area per open data initiatives, they can be used in place of the more generalized Commercial Reference Building inputs. Other input parameters that are not specified by the reference models or publically available datasets are given default values by engineering simulation tools like EnergyPlus. We acknowledge that different simulation programs (i.e., IES-VE, DOE-2, ESP-r) use varied default values as input parameters and rely on varied assumptions about thermal dynamics and energy processes to create an energy model. Theoretically, modeling the same building in two separate energy simulation programs would result in different results. However, this won't impact the overall performance of DUE-S, as the goal of first step in DUE-S is only to capture the baseline energy usage dynamics of each building based on the tradeoffs between the modeling objective and data availability (see Fig. 2). The second step of DUE-S and its residual network, discussed subsequently in Section 3.2, will further capture the remaining uncertainties about the generalized inputs and the impacts of urban context. As a result, DUE-S is agnostic to the simulation programs used to create the underlying baseline energy models

Once all data primary inputs for weather, geometry and non-geometric parameters are ready [82], multi-zone thermal models are created using OpenStudio, the EnergyPlus plugin for SketchUp. The 2.5-D massing models are constructed and divided into a pre-determined number of floors. Each floor is further divided into core and perimeter thermal zones, per ASHRAE 90.1 Annex G [83]. Each building is assigned an archetype (one type of U.S. Commercial Reference Building model) based on building age and use type. Finally, each individual building is simulated in EnergyPlus, resulting in an output of an IDF file and a time series of 15-min interval electricity use. We stress that the goal of these energy simulation models is to control their uncertainties from input parameters by analyzing the tradeoffs between the modeling objective and data availability, in order to capture the basic energy use dynamics of each building. The outputs (i.e., simulated energy use) of this step will then serve as inputs for the second step of the DUE-S modeling workflow: the residual network machine learning model.

3.2. Step 2: Residual network machine learning model

We theorize that the primary sources of uncertainty in building energy modeling arise from both the assumptions made in the energy simulation process as well as the "hidden" urban context impacts that are not captured by individual energy simulations. By using a deep learning algorithm to map simulated energy consumption to its actual building energy consumption, we expect this model will better account for the uncertainties in both the input parameters and assumptions and the urban context. More importantly, because interdependencies between buildings and the surrounding environment are dynamically coupled, the impact of the urban context takes a non-linear form and as a result can benefit from a deep network architecture that utilizes multiple hidden layers to capture this non-linearity.

In selecting a deep learning algorithm, we wanted to take full advantage of the spatial nature of the dense urban built environment. which a convolutional neural network and its locally-connected layers would be particularly efficient at handling. While time series forecasting is more commonly done using sequence models such as recurrent neural networks, past studies have employed CNNs and shown their ability to outperform these other models [76,77,84]. However, CNNs with very deep networks can also suffer from degradation (i.e., a saturation and rapid decline in training accuracy) [85]. This occurs when the information passed between convolution layers is not well estimated, which results in a "snowballing effect" of inaccurate predictions based on the poor performance of one layer. Residual networks (ResNets)-an extension of CNNs-apply convolutions in a similar manner to traditional CNNs but do not experience the high training losses associated with degradation. Both deep and shallow ResNets have been previously used in a spatiotemporal context to model crowd flow prediction [86] and have been shown to outperform traditional deep CNNs [87].

ResNets are a type of deep convolutional neural network made up of a series of residual blocks, as shown in Fig. 3a. An exhaustive explanation of residual networks and their architecture can be found in [85,88]. A residual block is made up of two convolutional layers and an identity mapping "skip" connection $a^{[l]}$ that can pass over layers to a deeper location in the neural network [88]. It is this "skip" connection in the ResNet architecture that makes it uniquely applicable to the context of urban energy modeling and the primary reason for choosing it in our initial testing of the DUE-S framework over other machine learning models. First order impacts of the urban context on energy use (e.g., shading) are captured by the traditional connections between layers in the ResNet. The "skip" connection enables the ResNet to learn the complex second-order impacts of surrounding buildings (e.g., thermal radiation, coupled occupant dynamics) of which may impact some buildings but "skip" buildings that are not impacted. As a result, the "skip" connection could enable the ResNet to capture the complex second-order effects of the urban context without forcing the model to over fit on a single layer and cause degradation (see Fig. 3b).

Specifically, as a^{II} skips through the first layer of the residual block, it passes through a linear function and a non-linear activation function:

$$a^{[l+1]} = g(Z^{[l+1]})$$

where **W** is the matrix of learned parameters for the ResNet, **a** is the activation for layer **l**, **b** is the bias vector, and **g** is the activation function. This process is then repeated through the second layer in the residual block, except for the newly introduced skip connection, which adds the original input information, a^{IU} , after the linear function but prior to the second non-linearity equation:

$$Z^{[l+2]} = W^{[l+2]}a^{[l+1]} + b^{[l+2]}$$

 $a^{[l+2]} = g(Z^{[l+2]} + a^{[l]})$

Batch normalization is often included as part of the ResNet architecture in order to more easily tune its hyperparameters and train the neural network [89]. The specific architecture of a residual block may vary depending on application, but we designed the residual block for this framework based on the results of [88] (Fig. 3b).

The overarching goal of the residual block is to learn the mappings between a^{II} and $a^{I^{I+2I}}$ while avoiding the issue of degradation. The output of the ResNet is the final prediction for building energy consumption, which, depending on the network architecture, could be at any time interval (e.g., hourly, daily, monthly, annually) or spatial scale (e.g., individual building, block, urban scale).

4. Case study

4.1. Study area and data inputs

To test the performance and feasibility of DUE-S, we draw data and information from an urban university campus located in California, USA. Because one of the goals of DUE-S is to capture how inter-building energy dynamics affect energy use, this site was selected based on the dense network of buildings on its campus. The study area (Fig. 4) is comprised of 22 university buildings, located on 4 adjacent blocks used primarily as offices and classrooms. We note that while the DUE-S model works for multiple energy types, this case study specifically analyzes electricity use as this data was only available and the most relevant for energy loads in the moderate climate of California, USA. A typical daily, monthly and yearly load profile for a building in the case study is provided in Appendix A. Its surrounding area contains additional buildings not included in this study and is paved for both car and pedestrian traffic with additional landscaping. As shown in Table 1, the data used to inform the baseline energy simulations was gathered through both publicly accessible datasets and the local university facilities department. We gathered three years of hourly historical weather data on the specific study area from the National Oceanic and Atmospheric Administration (NOAA) [80] and utilized this data for each year of simulation. To create the building geometries, we drew from a building GIS shapefile provided by Los Angeles County's LARIAC (LA Region Imagery Acquisition Consortium) program [90]. Since all the 22 buildings were built before 1980 and mainly consist of offices, the non-geometric properties for each building were defined based on the DOE Commercial Reference Building's "pre-1980s" construction and "medium office" use type. Additional interviews with a university facility manager defined the window-to-wall ratio (WWR) to range from 0.3 to 0.4. While BIM was available for some of the buildings on campus, we were more interested in evaluating the accuracy of DUE-S given fewer detailed inputs, as this would more likely be the circumstance facing an energy modeler on a city scale.

After simulating each of the 22 buildings for three years, each of their resulting 15-min interval time series of electricity consumption become the inputs to our residual network model. For our implementation of the ResNet, we shape our input data—the simulated data—into 1095 daily signals of 96 15-min time steps for all 22 buildings (Fig. 5). This input tensor with dimensions [22, 1095, 96] was then flattened to a 2D matrix of size [22, 105120] to feed into the



(simulated electricity use)





Fig. 4. Test bed study area for DUE-S.

Table 1

Datasets and data fields used for DUE-S baseline energy simulations	s.
---------------------------------------------------------------------	----

Use	Data source	Data type	Data field	
Building geometry	LA Region Imagery Acquisition Consortium (LARIAC) Client	Shapefile Interviews	Building area Building height Building elevation Num. of floors Average WWR Other specified non- geometric	
			parameters	
Weather data	NOAA	Time-series datasets	Historical hourly weather information	
Non-geometric parameters	DOE Commercial Reference Building (pre-1980s medium office)	Database	Occupancy schedules Building constructions HVAC type Heating/cooling loads	

ResNet. We then one-hot encode, or featurize, *day of the week* (7) and *month* (12) into our input space, increasing the number of input features to 41 while keeping the number of output features at 22 for a final flattened matrix of dimension [41, 105120]. We then split our data into training, development and testing sets. The training set, which is made up of 60% of the total data, is used for training several iterations of the ResNet—each of which use different combinations of hyperparameters (e.g., number of hidden layers, number of hidden units per layer,

Fig. 3. a (top). General ResNet architecture. Inputs to the model consist of periodic time series representing simulated energy use. These inputs are passed through a series of residual blocks that aim to learn the hidden relationships between simulated and metered energy use. b (bottom). Architecture for DUE-S ResNet residual block. Each residual block contains a series of convolution layers, batch normalizations and non-linear activation functions.

learning rate and activation function). The development set, which is made up of another 20% of the total data, is used to evaluate each of the models developed with the training set and to prevent overfitting when evaluating on the test set. Finally, the testing set, which contains the last 20% of the data, is used to evaluate the overall accuracy of the ResNet.

The training inputs were initially fed through a 1D convolutional layer that output 64 channels, determined through trial-and-error experimentation of three different channel sizes (32, 64, 128). These 64 channels were then passed through 8 residual blocks, each of which contain two 1D convolution layers, batch normalization and a leaky ReLU non-linear activation function with a coefficient of leakage α = 0.01, as shown in Fig. 3b. The number of residual blocks and activation function were chosen using an exhaustive trial-and-error tuning process. These 64 channels were passed through a final 1D convolution to output 22 channels, which were then fed through a fully connected logistic layer and a linear output layer, each containing 22 hidden units, to produce the final prediction for DUE-S. To optimize the network parameters, we used the Adam optimization algorithm to minimize MBE [91]. The ResNet's architecture was adjusted to evaluate the prediction accuracy of DUE-S at the individual building, block level and urban level, and the results are discussed in Section 5.

4.2. Validation metrics

(metered electricity use)

The performance of DUE-S depends on how reliable the simulation results match the actual electricity use. To evaluate the accuracy to which our model predicts the electricity use at the individual, block and urban scales, we compare the predicted electricity consumption to the metered electricity consumption at hourly, daily and monthly intervals using MBE, CV(RMSE) and MAPE. The first evaluation metric is the Mean Bias Error (MBE):

$$MBE = \frac{\sum_{i=1}^{n} Y_i - \widehat{Y}_i}{\sum_{i=1}^{n} Y_i}$$

In this equation, Y_i is the actual measurement of energy use and \hat{Y}_i is the simulated value; n is the number of measurements (this varies by time interval being studied with n = 1095 days, 36 months or 3 years in our case study). MBE is a non-dimensional metric for overall deviation, which can effectively reflect overestimation versus underestimation. It measures long-term model performance through analyzing the error between the simulated and measured energy use intensity.



Fig. 5. ResNet input space and dataset volume. After simulating each of the 22 buildings, the time series data was shaped into an input volume of 22 buildings \times 1095 days \times 96 15-min intervals of energy consumption. This volume was then flattened to become the 2-dimensional input space used in the ResNet model.

The second metric is the Coefficient of Variation (CV) of RMSE (Root Mean Square Error):

$$CV(RMSE) = \frac{\sqrt{\sum_{i=1}^{n} (Y_i - \hat{Y}_i)^2}}{(\sum_{i=1}^{n} (Y_i))/n}$$

The CV(RMSE) is determined by dividing the RMSE by the mean measured energy use intensity. It is not influenced by the compensation effect and can evaluate the variability of agreement between the simulated results and measured values over a period of time.

Finally, we evaluate the accuracy of the DUE-S case study using MAPE (Mean Absolute Percentage Error), which is a common metric used to evaluate the percentage error for forecasting applications:

$$MAPE = \left(\frac{1}{n} \sum_{i=1}^{n} \frac{|Y_i - \widehat{Y}_i|}{Y_i}\right) * 100$$

In previous studies, the time granularities of comparison include hourly [92], daily [93], monthly [94], yearly [19,95] and even aggregated value for the entire period [36]. However, the comparison is only limited for a few buildings in a small district due to the lack of available data. A comprehensive comparison at different spatial levels for validation is missing – a feature DUE-S is able to provide.

5. Results and discussion

After running the complete DUE-S framework for the case study buildings, we calculate the Mean Bias Error (MBE), Coefficient of Variation (CV) of RMSE and Mean Absolute Percentage Error (MAPE) per the equations discussed in Section 4.2. We note that the errors across our training, development and test data sets were consistent (e.g., monthly urban predictions had MAPE of 5.32%, 6.41% and 8.28% for training, development and testing datasets, respectively) and indicate that our ResNet model did not overfit to specific time period of the training data. The overall results are provided in Table 2 for all assessed spatial and temporal scenarios. To calculate the error results, we first aggregate the data temporally across all buildings to the desired scale for analysis (e.g., hourly, daily, monthly). Then, depending on the specified spatial scale, we aggregate the results, if necessary, to the block or urban scale.

In order to contextualize our results, we compare them to the ASHRAE Standard Guideline 14 [56], which defines metrics and guidelines for hourly and monthly building energy modeling for individual buildings. While ASHRAE currently does not have a standard to evaluate prediction accuracy at the block or urban scales, we are using individual building electricity use as a conservative proxy as other urban simulations have similarly done [48]. Based on the ASHRAE standard, we assess errors on the basis of metered electricity consumption in kilowatt-hours (kWh). Acceptable ranges for an energy simulation require an MBE of 10% for hourly intervals and 5% for monthly as well as a CV(RMSE) of 30% for hourly intervals and 15% for monthly. While MAPE is not commonly used to assess accuracy in energy simulation models, we include it in our analysis as it is often used for evaluating machine learning models. Additionally, different buildings might have variances in the magnitude of energy consumption, and as a result the MBE metric is susceptible to being dominated by larger buildings with higher electricity consumption. Utilizing the MAPE metric as an alternative metric allows for the normalization of errors across numerous buildings.

Overall the results shown in Table 2 demonstrate that DUE-S is able to more accurately predict electricity use as the spatial and temporal granularities increase. Compared to the ASHRAE Guidelines, DUE-S was able to more accurately predict electricity performance at the urban scale for hourly and monthly time intervals. We postulate that our model's success at these higher intervals is because aggregation reduces the variability in overall electricity consumption and creates a smoother consumption curve consistent with findings from previous sub-building [10] and residential [96] forecasting studies. Thus, by assessing electricity use on a wider spatial scale or at a larger time interval, the effects of buildings or blocks with worse individual predictions are dampened, indicating the impacts of urban context on to building energy use have been partially captured at higher spatial and temporal scales.

Fig. 6 shows the MAPE of DUE-S when predicting at the monthly interval and urban scale for the testing data. The trend indicates that the peak winter and summer months tend to perform worse than months in more temperate spring and fall seasons. One possible reason



Fig. 6. DUE-S performance in estimating urban scale electricity use at the monthly interval, evaluated with MAPE.



Fig. 7. DUE-S performance in estimating block scale electricity use at the monthly interval, evaluated with MAPE.

is that winter and summer months are generally when university vacation periods occur, and thus the use of buildings could be irregular and unpredictable. When we dissect monthly MAPE further at the block scale (Fig. 7), we see that while each block generally follows a similar trend in seasonal prediction with urban scale simulation, there is significant variability in the performance of each block. While the southwest block is unable to estimate electricity use nearly as well as the others, it is still able to follow a similar trend in seasonality which we believe shows that the ResNet is picking up on electricity use trends across large spatial regions. However, to fully understand the drivers for why this particular block of buildings is performing much worse, we further assess the monthly MAPE at the building scale (Fig. 8). Fig. 8 indicates that some buildings, similar to the block scale, are able to predict electricity use better or worse than others. Each row represents one of the 22 total buildings in the study area, whereas each column represents a single month of electricity prediction. Darker colors in the map represent worse MAPE scores and lighter colors represent more accurate results. As indicated earlier, the southwest block, which is comprised of the upper five buildings in the heatmap, contains five buildings that are not particularly accurate in their own electricity prediction, while each of the other three blocks contains buildings that were more accurate. One possible reason is the existence of missing data, which could significantly impact the learning of interdependencies and interactions of building energy use. Some of these buildings may perform better or worse than others, and we believe that



Fig. 9. Distribution of electricity use predictions by month.

may be also because the quality of metered data we were able to collect from the university varies in its completeness. For example, one of the worst performing buildings in the northwest block has 38% of missing values, while one of the best performing buildings from the southeast has 6% missing values. In turn, by assessing which buildings perform worse than others, we can focus future efforts to collect and clean electricity data for those particular buildings.

Finally, we examine the distribution of the monthly MAPE at the building scale in Fig. 9 to show how DUE-S performs for all buildings throughout the year. Among the twelve boxplots, nine have compact interquartile ranges (IQR) but long tails, indicating that during these months at the building scale MAPE deviates from each other but otherwise generally performs well. It is clear that there are buildings with much better estimates of electricity consumption than others. When averaged, however, the final result gives a much higher error (Table 2) due to the existence of outliers. It can also be seen that the boxplots are fallen into five regions (three are above the 0.00 line and two are below), demonstrating seasonality affects whether DUE-S is under or overestimating electricity use. The spring and fall months show a trend in underestimating electricity use while winter and summer ones tend to overestimate. During the months that DUE-S tends to underestimate electricity, the variance of MAPE is smaller than other months - indicating that the model performs better when providing a conservative estimate of electricity use.

It has been demonstrated that DUE-S is capable of accurately predicting building energy use at multiple temporal and spatial scales, which could be utilized for a variety of real-world urban energy efficiency applications, such as large-scale retrofit analysis. For example, if



Building-Scale DUE-S Predictions

Fig. 8. Heatmap of building scale predictions at the monthly interval, evaluated with MAPE. The heatmap is divided into four sections, each comprised of buildings located at each of the four blocks.

Table 2

Summary of DUE-S results at multiple temporal and spatial scales.

	MBE (%)			CV(RMSE) (CV(RMSE) (%)			MAPE (%)			
	Hourly	Daily	Monthly	Hourly	Daily	Monthly	Hourly	Daily	Monthly		
Building scale	38.8	27.9	17.3	46.0	31.3	27.9	49.6	43.1	43.8		
Block scale	27.1	18.2	14.7	40.6	25.1	18.5	21.4	18.3	16.9		
Urban scale	10.5	7.93	4.78	25.6	14.4	11.4	12.7	9.31	8.28		

an engineer was interested in understanding how changing the materiality of windows (e.g., from 6 mm Low-E glass to 12 mm Low-E glass) impacts energy use for numerous buildings in northwest block of our case study, they could utilize the DUE-S model as follows. First, they would run the DUE-S model with the baseline energy simulation to enable the residual network to learn the uncertainties associated with simplified assumptions made in the energy simulation process as well as the "hidden" urban context impacts. Next, they could take the baseline energy simulation and, holding all other input parameters constant, modify the glazing material properties of windows in the buildings receiving the potential retrofit. The resulting simulation results would show the basic energy dynamics of the proposed window retrofit. Then, the results from this retrofit simulation would become the inputs to the same residual network model created with the baseline energy simulation. By applying transfer learning [97] to the new input data, the pretrained network relating simulated and metered energy could also be used for the retrofit scenario. Because the only differentiation between the baseline and retrofit scenarios are the new modified window glazing inputs in the retrofit simulation, the difference in outputs from the residual network would help quantify the impact of the window retrofit on a single building in the northwest block, the rest of the block, and the ones around it.

Overall, the results of the case study underscore the value and merit of the integrated DUE-S approach, but they also demonstrate that significant amount of future work is necessary to further refine it at smaller spatial and temporal scales. Although it is common for deep learning models to use much more data than we had for this study, our ResNet was still able to pick up on the patterns presented in the available data. Furthermore, it was able to achieve results that demonstrate the potential of using deep learning to better predict energy use on multiple spatial and temporal scales by capturing some uncertainties in both modeling process and from the urban context. Besides working with a longer period of metered energy data, we believe that introducing more features to the input dataset, such as weather data or hours of operation, may help the ResNet pick up on additional patterns at the smaller spatial and temporal scales. Finally, with any data-driven model, its performance can only be as effective as the quality of its data. Because the inputs to the ResNet were the result of highly generalized energy simulations, we plan to further analyze the tradeoffs between modeling objectives and data availability to understand how the DUE-S results can be further improved. In the end, we aim to establish an integrated model that strikes the balance between extensibility and detail, as creating and maintaining comprehensive building energy simulation models for all buildings in a city is infeasible. This core objective of this paper is to propose a novel method for predicting building energy use at various spatial and temporal scales, and it represents the first key step to enable future work that explores the potential of using data-driven methods to create multiscale urban building energy models.

6. Limitations and future work

This paper aimed to take a first step in overcoming challenges faced in urban energy modeling and to test the feasibility of DUE-S: a generalized engineering urban energy simulation model integrated with a network-based machine learning model. Our goal was to understand whether an integrated model such as DUE-S could accurately characterize the energy performance of buildings at the individual building, block, and urban scale simultaneously. We evaluated our model through a case study of 22 co-located university buildings for 3 years and found that a residual neural network (ResNet) model was capable of predicting urban scale energy use on par with industry-accepted limits. However, future work is required to further validate the modeling approach on other case studies with diverse building stocks and urban morphologies to ensure the model is reliable for various scenarios. Moreover, our model was only able to successfully predict on the urban (campus) scale but demonstrated potential for achieving accuracy at other scales as well. The energy simulations relied on inputs provided by the Department of Energy's Commercial Reference Buildings and publicly available data, and specific details related to each individual building were not included in order to test the level of generalization a simulation could have to still be able to accurately predict energy use. Future work should aim to understand the level of specificity required of the energy simulations in order to improve prediction results within acceptable error ranges. Additionally, deep learning models, including ResNets, require a large number of observations in order to effectively train, develop and test their accuracy. Because one of the limitations of this study is the limited time period of provided data, we aim to train our ResNet on more observations in order to produce a model better capable of understanding the more complex patterns of building energy use. Future work also aims to analyze the performance of the ResNet in more detail by extracting the weights of each hidden layer to understand the correlations between consumption in adjacent buildings. Because the selected study area was a small cluster of university buildings, we were unable to determine its performance on a large number of densely clustered buildings of different sizes and use types. As such, we acknowledge that our case study is not fully representative of a large real city but nonetheless the results provide a high-level validation of how our proposed model would perform on a small and dense central business district (CBD) that is common in many parts of the United States. Finally, we aim to test DUE-S on a larger, more heterogeneous group of buildings in a different geographic area to further refine and improve the extensibility of the underlying framework, as well as implement DUE-S in real-world building and urban scale retrofit scenarios to support sustainable urban planning and operations.

7. Conclusion

The primary goal for this paper was to propose DUE-S: a data-driven modeling framework that integrates the computational power of a neural network model and interpretability of an engineering building energy simulation. The results of our case study analysis demonstrate the significant opportunity that exists to create more accurate multiscale urban energy models if emerging machine learning models were integrated into the current energy simulation workflow. However, in the current state, there is no generally accepted approach for urban energy modeling as there exists a significant amount of uncertainties in both buildings to be modeled and the surrounding environment. Current methods can only simulate buildings on an individual level, and fail to account for the energy dynamics and interdependencies resulting from urban context. The rapid development of new sensing technologies and smart city initiatives has led to an explosion data streams describing buildings and their urban environment, which in turn provide a significant opportunity to create more accurate multiscale urban energy models by integrating emerging data-driven machine learning models with an engineering simulation workflow to leverage the advantages of both. Specifically, we were able to demonstrate that our integrated DUE-S model achieves acceptable level of accuracy at the urban (campus) scale as per ASHRAE standards.

Overall, an integrated engineering and data-driven approach to urban energy performance modeling can yield a model generalizable for any city or dense building portfolio data and will be able to simulate energy consumption on an individual, block, and urban scale. By visualizing the energy use of buildings across a city, policymakers will have a better awareness of the effects of new citywide interventions (e.g., implementing an incentive program for window retrofits across the city). Designers and building operators will understand the effects of energy consumption on not only their building, but surrounding ones as well (e.g., adding reflective coating to minimize glare, retrofitting the HVAC system). And finally, as data is being increasingly used in the

Appendix A

planning of newer, smarter cities, this model can employ data from existing cities to optimize building energy use and help inform key decisions related to energy efficiency early on in the design process and as part of retrofit programming (e.g., developing planning policies related to land use and morphology, enacting a new policy to allow highrise development). In the end, as the world rapidly urbanizes modeling the energy performance of our urban buildings will undoubtedly unlock numerous opportunities that could potentially enhance the sustainability of our cities for years to come.

Acknowledgements

The material presented in this manuscript is based in part upon work supported by the UPS Foundation, Center for Integrated Facility Engineering (CIFE), a Terman Faculty Fellowship, and the US National Science Foundation (NSF) under Grants No. 1461549, 1642315. Any opinions, findings, and conclusions or recommendations expressed in this material are those of the author(s) and do not necessarily reflect the views of NSF.

Electricity use profiles based on metered electricity data for a single building in the campus study area. Profiles are shown in hourly intervals for peak summer and winter days, daily intervals for summer and winter months, and monthly intervals for a full year.



References

- United Nations. World's population increasingly urban with more than half living in urban areas; 2014. [Online]. Available: < http://www.un.org/en/development/ desa/news/population/world- urbanization-prospects-2014.html > .
- [2] United States Energy Information Administration. Annual energy outlook 2017; 2017. [Online]. Available: < https://www.eia.gov/outlooks/aeo/ > .
- [3] United Nations Environment Programme. Cities and buildings; 2013. [Online]. Available: < http://energies2050.org/wp-content/uploads/2013/09/2013-06-UNEP-Cities-and-buildings-activities_16-pages-GB.pdf > .
- [4] Breaking Energy. The top ten ways we waste energy in water and buildings; 2011. [Online]. Available: < http://breakingenergy.com/2011/07/26/the-top-ten-wayswe-waste-energy-and-water-in-buildings/ > .
- [5] San Francisco Dept. of Environment. San Francisco existing commercial buildings performance report; 2014. [Online]. Available: < https://sfenvironment.org/sites/ default/files/files/sfe_gb_ecb_performancereport.pdf > .
- [6] Pacific Gas & Electricity Company. Carbon footprint calculator assumptions. [Online]. Available: < https://pge.com/includes/docs/pdfs/about/environment/ calculator/assumptions.pdf > .
- [7] Pacific Gas & Electricity Company. Tariffs. [Online]. Available: < https://www.pge. com/tariffs/electric.shtml#COMMERCIAL > .
- [8] Singh KP, Gupta S, Rai P. Identifying pollution sources and predicting urban air quality using ensemble learning methods. Atmos Environ 2013;80:426–37.
- [9] Lv Y, Duan Y, Kang W, Li Z, Wang FY. Traffic flow prediction with big data: a deep learning approach. IEEE Trans Intell Transp Syst 2015;16(2):865–73.
- [10] Jain RK, Smith KM, Culligan PJ, Taylor JE. Forecasting energy consumption of

multi-family residential buildings using support vector regression: investigating the impact of temporal and spatial monitoring granularity on performance accuracy. Appl Energy 2014;123(Supplement C):168–78.

- [11] Pisello AL, Taylor JE, Xu X, Cotana F. Inter-building effect: simulating the impact of a network of buildings on the accuracy of building energy performance predictions. Build Environ 2012;58(Supplement C):37–45.
- [12] Samuelson H, Claussnitzer S, Goyal A, Chen Y, Romo-Castillo A. Parametric energy simulation in early design: high-rise residential buildings in urban contexts. Build Environ 2016;101:19–31.
- [13] Reinhart CF, Cerezo Davila C. Urban building energy modeling a review of a nascent field. Build Environ 2016;97:196–202.
- [14] Bueno B, Norford L, Pigeon G, Britter R. A resistance-capacitance network model for the analysis of the interactions between the energy performance of buildings and the urban climate. Build Environ 2012;54:116–25.
- [15] Bouyer J, Inard C, Musy M. Microclimatic coupling as a solution to improve
- building energy simulation in an urban context. Energy Build 2011;43(7):1549–59.
 [16] Chen Y, Hong T, Piette MA. Automatic generation and simulation of urban building energy models based on city datasets for city-scale building retrofit analysis. Appl Energy 2017;205:323–35.
- [17] Liu J, Heidarinejad M, Gracik S, Srebric J. The impact of exterior surface convective heat transfer coefficients on the building energy consumption in urban neighborhoods with different plan area densities. Energy Build 2015;86(Supplement C):449–63.
- [18] Toparlar Y, Blocken B, Vos P, Van Heijst GJF, Janssen WD, Van Hooff T. CFD simulation and validation of urban microclimate: a case study for Bergpolder Zuid, Rotterdam, no. August; 2014. p. 1–21.
- [19] Strzalka A, Bogdahn J, Coors V, Eicker U. 3D City modeling for urban scale heating energy demand forecasting 3D City modeling for urban scale heating energy demand forecasting, no. August 2011; 2011. p. 37–41.
- [20] Huber J, Nytsch-geusen C. Development of modeling and simulation strategies for large-scale urban districts weather therm. Build Simul Model Plant Simul 2011:14–16.
- [21] Orehounig K, Evins R, Dorer V. Integration of decentralized energy systems in neighbourhoods using the energy hub approach. Appl Energy 2015;154(Supplement C):277–89.
- [22] Han Y, Taylor JE, Pisello AL. Exploring mutual shading and mutual reflection interbuilding effects on building energy performance. Appl Energy 2017;185(Part 2):1556–64.
- [23] Pisello AL, Castaldo VL, Taylor JE, Cotana F. The impact of natural ventilation on building energy requirement at inter-building scale. Energy Build 2016;127(Supplement C):870–83.
- [24] Pisello AL, Castaldo VL, Taylor JE, Cotana F. Expanding Inter-Building Effect modeling to examine primary energy for lighting. Energy Build 2014;76(Supplement C):513–23.
- [25] Allegrini J, Orehounig K, Mavromatidis G, Ruesch F, Dorer V, Evins R. A review of modelling approaches and tools for the simulation of district-scale energy systems. Renew Sustain Energy Rev 2015;52:1391–404.
- [26] Dall'O' G, Galante A, Torri M. A methodology for the energy performance classification of residential building stock on an urban scale. Energy Build 2012;48(Supplement C):211–9.
- [27] Mastrucci A, Baume O, Stazi F, Salvucci S, Leopold U, Bianche VB, A GIS-based approach to estimate energy savings and indoor thermal comfort for urban housing stock retrofitting. Dipartimento di Ingegneria Civile, Edile e Architettura – Università Politecnica delle Marche, Resource Centre for Environmental Technologies; 2013.
- [28] Alhamwi A, Medjroubi W, Vogt T, Agert C. GIS-based urban energy systems models and tools: introducing a model for the optimisation of flexibilisation technologies in urban areas. Appl Energy 2017;191(Supplement C):1–9.
- [29] Kaden R, Kolbe T. City-wide total energy demand estimation of buildings using semantic 3D city models and statistical data 2013;II-2/W1.
- [30] Cerezo Davila C, Dogan T, Reinhart CF. Towards standardized building properties template files for early design energy model generation. 2014 ASHRAE/IBPSA-USA Build Simul Conf; 2014.
- [31] Dogan T, Reinhart C. Automated conversion of architectural massing models into thermal 'Shoebox' Models. Cambridge MA, USA: Massachusetts Institute of Technology; 2006. p. 3745–52.
- [32] Autodesk white paper. Streamlining energy analysis of existing buildings with rapid energy modeling; 2011.
- [33] Jaffal I, Inard C, Ghiaus C. Fast method to predict building heating demand based on the design of experiments. Energy Build 2009;41(6):669–77.
- [34] Ghiassi N, Tahmasebi F, Mahdavi A. Harnessing buildings' operational diversity in a computational framework for high-resolution urban energy modeling. Build Simul 2017;10(6):1005–21.
- [35] Robinson D et al. CITYSIM: comprehensive micro-simulation of resource flows for sustainable urban planning. Int IBPSA Conf 2009:1083–90.
- [36] Nouvel R, Schulte C, Eicker U, Pietruschka D, Coors V. Citygml-based 3D city model for energy diagnostics and urban energy policy support HFT – zafh. net, Stuttgart, Germany HFT – geoinformatic, Stuttgart, Germany Description of the integrated. p. 218–25.
- [37] Kim EJ, Plessis G, Hubert JL, Roux JJ. Urban energy simulation: simplification and reduction of building envelope models. Energy Build 2014;84:193–202.
- [38] Yang X, Zhao L, Bruse M, Meng Q. An integrated simulation method for building energy performance assessment in urban environments. Energy Build 2012;54:243–51.
- [39] Morakinyo TE, Dahanayake KWDKC, Adegun OB, Balogun AA. Modelling the effect of tree-shading on summer indoor and outdoor thermal condition of two similar

buildings in a Nigerian university. Energy Build 2016;130(Supplement C):721-32.

- [40] Reinhart CF, Dogan T, Jakubiec JA, Rakha T, Sang A. Umi an urban simulation environment for building energy use, daylighting and walkability. Proc BS2013 13th Conf Int Build Perform Simul Assoc; 2013. p. 476–83.
- [41] Mauree D, Coccolo S, Kaempf J, Scartezzini J-L. Multi-scale modelling to evaluate building energy consumption at the neighbourhood scale. PLoS One 2017.
- [42] Clarke JA, et al. Simulation-assisted control in building energy management systems. Energy Build 2002;34(9):933–40.
- [43] Howard B, Parshall L, Thompson J, Hammer S, Dickinson J, Modi V. Spatial distribution of urban building energy consumption by end use. Energy Build 2012;45:141–51.
- [44] Yang Z, Becerik-Gerber B. A model calibration framework for simultaneous multilevel building energy simulation. Appl Energy 2015;149:415–31.
- [45] Bernal W, Behl M, Nghiem T, Mangharam R. Campus-wide integrated building energy simulation. BS 2015 – 14th Int IBPSA Conf 2015;December:2765–72.
- [46] Koene FGH, Bakker LG, Lanceta D, Narmsara S. Simplified building model of districts HTFD, TNO Delft, The Netherlands SOLINTEL M & P, Madrid, Spain STEP 1 REPRESENTING A SINGLE BUILDING BY A TWO-MASS MODEL. p. 152–9.
- [47] Firth SK, Lomas KJ. Investigating Co2 emission reductions in existing urban housing using a community domestic energy model. Build Simul 2009:2098–105.
- [48] Sokol J, Cerezo Davila C, Reinhart CF. Validation of a Bayesian-based method for defining residential archetypes in urban building energy models. Energy Build 2017;134:11–24.
- [49] Booth AT, Choudhary R, Spiegelhalter DJ. A hierarchical Bayesian framework for calibrating micro-level models with macro-level data. J Build Perform Simul 2013;6(4).
- [50] Orehounig K, Mavromatidis G, Evins R, Dorer V, Carmeliet J. Predicting energy consumption of a neighbourhood using building performance simulations. Proc BSO14; 2014.
- [51] Ahmad M, Culp CH. Uncalibrated building energy simulation modeling results. HVAC&R Res Oct. 2006;12(4):1141–55.
- [52] Li N, Yang Z, Becerik-Gerber B, Tang C, Chen N. Why is the reliability of building simulation limited as a tool for evaluating energy conservation measures?. Appl Energy 2015;159(Supplement C):196–205.
- [53] Manfren M, Aste N, Moshksar R. Calibration and uncertainty analysis for computer models – a meta-model based approach for integrated building energy simulation. Appl Energy 2013;103(Supplement C):627–41.
- [54] Monfet D et al. Calibration of a building energy model using measured data. ASHRAE Trans 2009;115.1.
- [55] Parker J, Cropper P, Shao L. A calibrated whole building simulation approach to assessing retrofit options for Birmingham airport. In: Proceedings of the IBPSA—England; 2012.
- [56] ASHRAE. ASHRAE guideline 14-2002: measurement of energy and demand savings. ASHRAE, vol. Atlanta, G; 2002.
- [57] US Department of Energy. M&V guidelines: measurement and verification for federal energy projects, vol. Version 3; 2008.
- [58] Heo Y, Choudhary R, Augenbroe GA. Calibration of building energy models for retrofit analysis under uncertainty. Energy Build 2012;47:550–60.
- [59] Reddy T, Maor I, Panjapornpon C. Calibrating detailed building energy simulation programs with measured data – Part I: general methodology. HVAC R Res 2007:13:221–41.
- [60] Magnier L, Haghighat F. Multiobjective optimization of building design using TRNSYS simulations, genetic algorithm, and Artificial Neural Network. Build Environ 2010;45(3):739–46.
- [61] Karunakaran R, Iniyan S, Goic R. Energy efficient fuzzy based combined variable refrigerant volume and variable air volume air conditioning system for buildings. Appl Energy 2010;87(4):1158–75.
- [62] Sanyal J, New J, Edwards R. Supercomputer assisted generation of machine learning agents for the calibration of building energy models. In: Proceedings of the conference on extreme science and engineering discovery environment: gateway to discovery; 2013.
- [63] Zhao H, Magoulès F. A review on the prediction of building energy consumption. Renew Sustain Energy Rev 2012;16(6):3586–92.
- [64] Asadi S, Amiri SS, Mottahedi M. On the development of multi-linear regression analysis to assess energy consumption in the early stages of building design. Energy Build 2014;85:246–55.
- [65] Dong B, Cao C, Lee SE. Applying support vector machines to predict building energy consumption in tropical region. Energy Build 2005;37(5):545–53.
- [66] Yu Z, Haghighat F, Fung BCM, Yoshino H. A decision tree method for building energy demand modeling. Energy Build 2010;42(10):1637–46.
- [67] Ekici BB, Aksoy UT. Prediction of building energy consumption by using artificial neural networks. Adv Eng Softw 2009;40(5):356–62.
- [68] Tso GKF, Yau KKW. Predicting electricity energy consumption: a comparison of regression analysis, decision tree and neural networks. Energy 2007;32(9):1761–8.
- [69] Fonseca JA, Schlueter A. Integrated model for characterization of spatiotemporal building energy consumption patterns in neighborhoods and city districts. Appl Energy 2015;142:247–65.
- [70] Li Z, Han Y, Xu P. Methods for benchmarking building energy consumption against its past or intended performance: an overview. Appl Energy 2014;124(Supplement C):325–34.
- [71] Wong SL, Wan KKW, Lam TNT. Artificial neural networks for energy analysis of office buildings with daylighting. Appl Energy 2010;87(2):551–7.
- [72] Hong S-M, Paterson G, Mumovic D, Steadman P. Improved benchmarking comparability for energy consumption in schools. Build Res Inf Jan. 2014;42(1):47–61.
- [73] Kontokosta CE, Tull C. A data-driven predictive model of city-scale energy use in buildings. Appl Energy 2017;197:303–17.

- [74] Bottou L, Large-scale machine learning with stochastic gradient descent. In: Proceedings of COMPSTAT'2010: 19th international conference on computational statistics, Paris, France, August 22–27, 2010; 2010. p. 177–86.
- [75] Ghaderi A, Sanandaji BM, Ghaderi F. Deep forecast: deep learning-based spatiotemporal forecasting 2017;ii.
- [76] Dai Z, He Z, Wang Y. Learning traffic as images: a deep convolution neural network for large – scale transportation network speed prediction. Sensors MDPI 2017:1–16.
- [77] Amarasinghe K, Marino DL, Manic M. Deep neural networks for energy load forecasting. 2017 IEEE 26th Int Symp Ind Electron; 2017. p. 1483–8.
- [78] Coakley D et al. Calibration of a detailed BES model to measured data using an evidence-based analytical optimisation approach. In: Proceedings of building simulation 2011; 2011.
- [79] Crawley LKL, Crawley Drury B, Hand Jon W. Improving the weather information available to simulation programs. Drury B. Crawley U.S. Department of Energy Linda K. Lawrie U.S. Army Construction Engineering Research Laboratory Champaign, Illinois 61821 USA. Proc Build Simul '99, vol. 2, no. Esru; 1999.
- [80] NOAA. NOAA baseline climatological dataset hourly weather station temperature and precipitation data. [Online]. Available: < https://www.ncdc.noaa.gov/ >.
- [81] Deru M et al. U.S. Department of Energy commercial reference building models of the national building stock. Publ., no. February 2011; 2011. p. 1–118.
- [82] Fabrizio E, Monetti V. Methodologies and advancements in the calibration of building energy models. Energies 2015;8(4):2548–74.
- [83] Calm JM et al. ASHRAE STANDARD energy standard for buildings except low-rise residential buildings. Society 2007;8400:404–636.
- [84] Borovykh A, Bohte S, Oosterlee CW. Conditional time series forecasting with convolutional neural networks; 2017. p. 1–19.

- [85] Wu S, Zhong S, Liu Y. Deep residual learning for image recognition. Multimed Tools Appl 2017:1–17.
- [86] Zhang J, Zheng Y, Qi D. Deep spatio-temporal residual networks for citywide crowd flows prediction; 2016.
- [87] Zhang J, Zheng Y, Qi D, Li R, Yi X. DNN-based prediction model for spatial-temporal data. Sigspatial 2016:1–4.
- [88] He K, Zhang X, Ren S, Sun J. Identity mappings in deep residual networks. Lect Notes Comput Sci (including Subser Lect Notes Artif Intell Lect Notes Bioinformatics) 2016;9908 LNCS:630–45.
- [89] Ioffe S, Szegedy C. Batch normalization: accelerating deep network training by reducing internal covariate shift; 2015.
- [90] Los Angeles County GIS Data Portal. Countywide building outlines 2014; 2014.
 [91] Kingma DP, Ba J. Adam: a method for stochastic optimization. CoRR 2014;abs/ 1412.6.
- [92] Heiple S, Sailor DJ. Using building energy simulation and geospatial modeling techniques to determine high resolution building sector energy consumption profiles. Energy Build 2008;40(8):1426–36.
- [93] Lauster M, Teichmann J, Fuchs M, Streblow R, Mueller D. Low order thermal network models for dynamic simulations of buildings on city district scale. Build Environ 2014;73.
- [94] Bahu JM, Koch A, Kremers E, Murshed SM. Towards a 3D spatial urban energy modelling approach. ISPRS Ann Photogramm Rem Sens Spat Inf Sci 2013;II-2/ W1(November):33–41.
- [95] Mata É, Sasic Kalagasidis A, Johnsson F. Building-stock aggregation through archetype buildings: France, Germany, Spain and the UK. Build Environ 2014;81:270–82.